Author's response to reviews

Title: Feasibility of identifying families for genetic studies of birth defects using the National Health Interview Survey

Authors:

Dr Diego F Wyszynski (dfw@bu.edu)
Vikki G Nolan (vnolan@bu.edu)

Version: 5 Date: 11 May 2004

We removed all unnecessary capitalisation from the title as suggested.